\begin{table}%t1
%\centering
\par
\caption{\label{tbl-1}Target list.}
\begin{tabular}{rlcccllc}
\hline \hline \noalign{\smallskip}
 & \multicolumn{1}{c}{Source} & RA & Dec  & $V_{\rm lsr}$ & 
\multicolumn{1}{c}{Obs.\tablefootmark{a}} & \multicolumn{1}{c}{Molecules\tablefootmark{b}} & References\tablefootmark{c}\\
\multicolumn{1}{c}{\#} & & \multicolumn{1}{c}{(J2000.0)} & 
\multicolumn{1}{c}{(J2000.0)} & (\kms) \\
\hline 
1 &L1355 & 02:53:12.1 & $+$68:55:52 & $-3.7$ & $m, e$   & \amm\  & 14\\ 
2 &NH3SRC8 & 03:25:26.3 & $+$30:45:05 & $+4.1$ & $m, e, n $ & \amm, \nnhp  & 5\\
3 &NH3SRC28 & 03:27:26.4 & $+$29:51:08 & $+5.6$ & $e$   & \amm\  & 5\\
4 &NH3SRC36 & 03:27:55.9 & $+$30:06:18 & $+4.7$ & $m$   & \amm\ & 5\\
5 &NH3SRC91 & 03:30:15.2 & $+$30:23:39 & $+5.9$ & $m$ &\amm & 5\\
6 &NH3SRC95 & 03:30:32.1 & $+$30:26:19 & $+6.1$ & $m$   & \amm, \hcccn & 5\\
7 &NH3SRC97 & 03:30:50.5 & $+$30:49:17 & $+7.7$ & $e$   & \amm & 5\\
8 &L1489    & 04:04:49.0 & $+$26:18:42 & $+6.7$ & $e, n$ & \amm,  \hcccn, \nnhp & 1, 4\\ 
9 &L1498    & 04:10:51.4 & $+$25:09:58 & $+7.8$ & $e, n$ & \amm, \hcccn, \nnhp & 1, 3, 4, 6\\
10 &L1495    & 04:14:08.2 & $+$28:08:16 & $+6.8$ & $e, n$ & \amm, \hcccn, \nnhp & 1, 6\\
11 &L1521F   & 04:28:39.8 & $+$26:51:35 & $+6.5$ & $e$ & \amm, \hcccn & 4, 6, 10\\ 
12 &L1524    & 04:29:22.7 & $+$24:34:58 & $+6.5$ & $e, n$ & \amm, \hcccn, \nnhp & 4, 8, 10\\ 
13 &B217S-W   & 04:27:46.4 & $+$26:17:52 & $+7.0$ & $e$ & \amm, \hcccn & 1, 4, 12\\ 
14 &L1400K    & 04:30:52.0 & $+$54:51:55 & $+3.3$ & $e, n$ & \amm, \hcccn & 1, 3, 4, 6\\ 
15 &TMC-2     & 04:32:48.6 & $+$24:25:12 & $+6.2$ & $e, n$ & \amm, \hcccn, \nnhp & 1\\
16 &L1536     & 04:33:23.5 & $+$22:42:46 & $+5.6$ & $e, n$ & \amm, \hcccn, \nnhp & 1, 4\\ 
17 &CB22      & 04:40:33.1 & $+$29:55:04 & $+6.0$ & $e$ & \amm, \hcccn & 4\\ 
18 &TMC-1C    & 04:41:38.8 & $+$26:00:42 & $+5.2$ & $m,e,n$ & \amm, \hcccn, \nnhp & 1, 2, 3, 4\\
19 &TMC-1/HC3N& 04:41:41.8 & $+$25:41:42 & $+5.9$ & $m$ & \amm, \hcccn & 9, 11\\ 
20 &CB23      & 04:43:27.7 & $+$29:39:11 & $+6.0$ & $e, n$ & \amm, \hcccn, \nnhp & 4, 6\\
21 &L1517B    & 04:55:18.8 & $+$30:38:04 & $+5.8$ & $e, n$ & \amm, \hcccn, \nnhp & 2, 4, 6\\
22 &L1512     & 05:04:09.6 & $+$32:43:09 & $+7.1$ & $m, e, n$ & \amm, \hcccn, \nnhp & 1, 2, 4, 6, 8\\ 
23 &L1544     & 05:04:16.5 & $+$25:10:48 & $+7.2$ & $e, n$ & \amm, \hcccn & 3, 4, 6, 8, 9\\
24 &L134A     & 15:53:36.2 & $-$04:35:26 & $+2.7$ & $e$ & \amm & 4, 6, 8\\ 
25 &L183      & 15:54:06.4 & $-$02:52:23 & $+2.4$ & $m, e, n$ & \amm, \hcccn, \nnhp & 4, 6, 7, 8\\ 
26 &L43       & 16:34:35.0 & $-$15:46:36 & $+0.7$ & $e, n$ & \amm, \hcccn, \nnhp & 4 \\ 
27 &L260-NH3  & 16:47:06.6 & $-09$:35:21 & $+3.5$ & $e, n$ & \amm, \hcccn, \nnhp & 1, 8, 12\\ 
28 &L234A     & 16:48:06.8 & $-$10:51:48 & $+2.9$ & $e$ & \amm, \hcccn & 4, 8\\ 
29 &L234S-E   & 16:48:08.6 & $-$10:57:25 & $+3.2$ & $e$ & \amm & 4\\
30 &L63       & 16:50:15.4 & $-$18:06:06 & $+5.7$ & $e$ & \amm, \hcccn & 4, 8\\ 
31 &CB68      & 16:57:19.4 & $-$16:09:25 & $+5.2$ & $e$ & \amm, \hcccn & 4 \\ 
32 &L492      & 18:15:49.6 & $-$03:46:14 & $+7.7$ & $e$ & \amm, \hcccn & 6\\ 
33 &L778      & 19:26:32.5 & $+$23:58:42 & $+9.9$ & $m$ & \amm & 1, 4\\ 
34 &B335      & 19:37:01.3 & $+$07:34:07 & $+8.4$ & $e$ & \amm, \hcccn & 4, 8\\ 
35 &L694-2    & 19:41:04.4 & $+$10:57:02 & $+9.5$ & $m, e$ & \amm & 6, 13\\ 
36 &L1049-2   & 20:41:30.5 & $+$57:33:44 & $+0.1$ & $e$ & \amm & 15 \\ 
37 &L1251T3   & 22:29:50.3 & $+$75:13:25 & $-3.8$ & $e, n$ & \amm, \hcccn, \nnhp & 4 \\ 
38 &L1251T2   & 22:30:31.2 & $+$75:13:38 & $-4.0$ & $e, n$ & \amm, \hcccn, \nnhp & 4 \\ 
39 &L1251T1   & 22:31:02.3 & $+$75:13:39 & $-4.2$ & $e, n$ & \amm, \hcccn, \nnhp & 4 \\
40 &L1251C    & 22:35:53.6 & $+$75:18:55 & $-4.7$ & $e, n$ & \amm, \hcccn, \nnhp & 16,17 \\ 
41 &L1197     & 22:37:02.1 & $+$58:57:22 & $-3.1$ & $e, n$ & \amm, \hcccn, \nnhp & 6,13 \\ 
\hline
\end{tabular}
\tablefoot{\tablefoottext{a}{Observations: $m$~-- Medicina 32-m, $e$~-- Effelsberg 100-m, and $n$~-- Nobeyama 45-m radio telescopes.} 
\tablefoottext{b}{Detected transitions:  NH$_3$ $(J,K) = (1,1)$,  HC$_3$N $J = 2-1$, N$_2$H$^+$ $J = 1-0$.} 
\tablefoottext{c}{References: (1)~Benson \& Myers (\cite{Benson1989}); (2)~Myers \etal\ (\cite{Myers1983});  (3)~Howe \etal\ (\cite{Howe1994}); (4)~Jijina \etal\ (\cite{Jijina1999});   (5)~Rosolowsky \etal\ (\cite{Rosolowsky2008}); (6)~Crapsi \etal\ (\cite{Crapsi2005}); (7)~Pagani \etal\ (\cite{Pagani2009}); (8)~Fuller \& Myers (\cite{Fuller1993});   (9)~Churchwell \etal\ (\cite{Churchwell1984}); (10)~Codella \etal\ (\cite{Codella1997}); 
(11)~Benson \& Myers (\cite{Benson1983}); (12)~Motte \& Andre (\cite{Motte2001});   (13)~Lee \etal\ (\cite{Lee2007}); (14)~Park \etal\ (\cite{Park2004}); (15)~Lee \etal\ (\cite{Lee1999}); (16)~Caselli  \etal\ (\cite{Caselli2002}); (17)~Walsh \etal\ (\cite{Walsh2004}).} }
\end{table}